Author's response to reviews

Title: Factors associated with postpartum hemorrhage maternal death in referral hospitals in Senegal and Mali: a cross-sectional epidemiological survey.

Authors:

Julie TORT (julie.tort@gmail.com)
Patrick ROZENBERG (prozenberg@chi-poissy-st-germain.fr)
Mamadou TRAORE (mtraore133@gmail.com)
Pierre FOURNIER (pierre.fournier@umontreal.ca)
Alexandre DUMONT (alexandre.dumont@ird.fr)

Version: 6 Date: 9 June 2015

Author's response to reviews: see over
Dear Editor in chief,

Thank you for your review and comments. Please find enclose our comments of the revised manuscript entitled: “Post-partum hemorrhage in referral hospitals in Senegal and Mali: Factors associated with maternal death” for re-submission to BMC Pregnancy & Childbirth.

1) The authors have added sentences on lines 119-120 regarding the location of most births in these countries, as requested. However, they have not included a reference(s) for the figures they state (55% home and 45% community health care centers).

   We revised this sentence including the most recent national statistics as “According to available statistics in Mali and Senegal, among the remaining 90% of deliveries, approximately 50% occurred at home and 50% in community health care centers in Mali and approximately 32% occurred at home and 68% in community health care centers in Senegal.” (line 119).

2) Can the authors confirm that they edited throughout for correct English usage?
Yes, we confirm that we edited throughout for correct English, and that we change the style of the article, but not the content. All modifications have been highlighted in yellow.

Sincerely Yours,

Julie TORT, PhD candidate.
UMR 216, Faculté de Pharmacie, Laboratoire de Parasitologie, 4 Avenue de l’Observatoire, 75270 Paris Cedex 6, France.
Tel.: 33 (0)1 53 73 96 57
julie.tort@gmail.com